Mauna Loa Solar Observatory Observer's Log _____ Tue Apr 7 17:00:24 GMT 2015 Year: 15 Doy: 097 Observer: stueben WEATHER COMMENT: stueben: Tue Apr 7 17:00:33 GMT 2015 Clear, wind southeast 5 mph, temp 38F. end PSPT COMMENT BY stueben: Tue Apr 7 17:16:46 GMT 2015 Pspt started. end Tue Apr 07 17:24:23 GMT 2015 COMP Start Patrol Tue Apr 07 17:38:42 GMT 2015 KCOR Start Synoptic Patrol **GENERAL PROBLEM COMMENT BY stueben** : Tue Apr 7 18:28:20 GMT 2015 No internet access via the desktop or wireless access with the laptop. Instrument computers seem OK except the Shabar lapt op. Is this related to yesterdays router changes? The Network Center can't find the problem for the desktop and for the la ptop wireless it says that the DNS server is not responding. ____end_ Tue Apr 07 18:40:00 GMT 2015 KCOR End Patrol Tue Apr 07 18:40:01 GMT 2015 KCOR Start Calibration script: c:\kcor\mlso-calibration22deg-20150323.ini Tue Apr 07 18:57:05 GMT 2015 KCOR End Calibration Script Tue Apr 07 18:57:21 GMT 2015 KCOR Start Synoptic Patrol Tue Apr 07 18:57:22 GMT 2015 KCOR Start Synoptic Patrol Tue Apr 07 19:13:07 GMT 2015 KCOR End Patrol Tue Apr 07 19:13:48 GMT 2015 CoMP Paused for clouds WEATHER COMMENT: stueben: Tue Apr 7 19:15:35 GMT 2015 Orographic clouds are degrading images, comp and kcor idled. end **GENERAL PROBLEM COMMENT BY stueben** : Tue Apr 7 19:22:46 GMT 2015 Internet access was apparently broken by an unscheduled change to the router/mlsoserver. Ben reverted it and I now have in ternet access again. end Tue Apr 07 19:44:58 GMT 2015 CoMP Restarted from pause Tue Apr 07 19:44:59 GMT 2015 COMP End Patrol WEATHER COMMENT: stueben: Tue Apr 7 19:45:36 GMT 2015 Blowing fog, instruments idled and domes closed. end WEATHER COMMENT: stueben: Wed Apr 8 01:26:13 GMT 2015 Fog and intermittant rain.

____end____